Author's response to reviews

Title: Delivery of maternal health care in Indigenous primary care services: baseline data for an ongoing quality improvement initiative

Authors:

Alice R Rumbold (alice.rumbold@adelaide.edu.au)
Ross S Bailie (ross.bailie@menzies.edu.au)
Damin Si (damin.si@menzies.edu.au)
Michelle C Dowden (michelldowden@ngalkanbuy.com.au)
Catherine M Kennedy (cath.kennedy@bigpond.com)
Rhonda J Cox (R.Cox@curtin.edu.au)
Lynette O’Donoghue (Lynette.O’Donoghue@menzies.edu.au)
Helen E Liddle (helen.liddle@menzies.edu.au)
Ru K Kwedza (Ru_Kwedza@health.qld.gov.au)
Sandra C Thompson (S.Thompson@curtin.edu.au)
Hugh P Burke (maarimaact@bigpond.com)
Alex DH Brown (alex.brown@bakeridi.edu.au)
Tarun Weeramanthri (Tarun.Weeramanthri@health.wa.gov.au)
Christine M Connors (Christine.connors@nt.gov.au)

Version: 4 Date: 3 March 2011

Author's response to reviews:

Response to comments from the Production team:

Major revisions

Figures: All figures must be cited; figures not cited will be excluded from the final manuscript.

Response: Please remove figure 1 from the manuscript. We included this figure for the editors to consider in the response to reviewers dated 22nd December 2010, however we feel that this map is not very informative and that the text in the methods sufficiently describes the wide distribution and very remote locations of many of the participating services.

Figure 1: please provide a short title for figure 1. The legend and title should be part of the manuscript file after the reference list.

Response: As above – please remove this figure from the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details
Response: All affiliations have been amended to include the country of each author.

Abbreviations: Please format these in to a paragraph.
Response: The abbreviations have been reformatted into a paragraph.

References: Please do not include DOI numbers in your reference list
Response: The DOI has been removed.